Author’s response to reviews

Title: Metabolic syndrome in hypertensive women in the age of menopause: A case study on data from General Practice Electronic Health Records

Authors:

Sefket Sabanovic (obiteljska6@gmail.com)
Ljiljana Majnaric (Ljiljana.majnaric@gmail.com)
Frantisek Babik (frantisek.babic@tuke.sk)
Michal Vadovsky (michal.vadovsky@tuke.sk)
Jan Paralic (jan.paralic@tuke.sk)
Aleksandar Vcev (aleksandar.vcev@mefos.hr)
Andreas Holzinger (a.holzinger@tugraz.at)

Version: 2 Date: 22 Feb 2018

Author’s response to reviews:

1. Please make sure that there is one 'Ethics approval and consent to participate' statement, not two separate ones.

   Answer: Done

2. Please add a section ‘Figures legend’ (usually after the references) where you list the following information for each figure:

   - Figure name
   - Description of a figure

   Answer: Done

3. We note that one of the authors is part of the Editorial Board for the journal. For the transparency of peer review, may you please state his role in the competing interests statement.

   Answer: Done
4. On uploading your revisions, please remove any tracked changes or highlighting and include only a single clean copy.

Answer: Done